Minocycline for negative symptoms of schizophrenia and possible mechanistic actions: the BeneMin RCT

Bill Deakin, ^{1,2,3}* John Suckling, ^{4,5} Paola Dazzan, ⁶ Eileen Joyce, ⁷ Stephen M Lawrie, ⁸ Rachel Upthegrove, ⁹ Nusrat Husain, ^{10,11} Imran B Chaudhry, ^{11,12} Graham Dunn, ^{13†} Peter B Jones, ^{4,5} Danuta Lisiecka-Ford, ⁴ Shôn Lewis, ^{1,2,3} Thomas RE Barnes, ¹⁴ Steven CR Williams, ¹⁵ Carmine M Pariante, ¹⁶ Emma Knox, ¹ Richard J Drake, ^{2,10} Richard Smallman and Nicholas M Barnes ¹⁷ on behalf of the BeneMin study team

¹Neuroscience and Psychiatry Unit, University of Manchester, Manchester, UK

²Greater Manchester Mental Health NHS Foundation Trust, Manchester, UK

³Manchester Academic Health Science Centre, Core Technology Facility, University of Manchester, Manchester, UK

⁴Brain Mapping Unit, Department of Psychiatry, University of Cambridge, Cambridge, UK

⁵Cambridge and Peterborough NHS Foundation Trust, Cambridge, UK

⁶Department of Psychosis Studies, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

⁷Sobell Department of Motor Neuroscience and Movement Disorders, Institute of Neurology, University College London, London, UK

⁸Division of Psychiatry, University of Edinburgh, Edinburgh, UK

⁹Institute for Mental Health, University of Birmingham, Birmingham, UK

¹⁰Division of Psychology and Mental Health, School of Health Sciences, Faculty of Biology, Medicine and Health, University of Manchester, Manchester, UK

¹¹Lancashire Care Early Intervention Service, Accrington, UK

¹²Division of Neuroscience and Experimental Psychology, School of Biological Sciences, Faculty of Biology, Medicine and Health, University of Manchester, Manchester, UK

¹³Division of Population Health, Health Services Research and Primary Care, School of Health Sciences, Faculty of Biology, Medicine and Health, University of Manchester, Manchester, UK

¹⁴Centre for Psychiatry, Department of Medicine, Imperial College London, London, UK

¹⁵Department of Neuroimaging, Institute of Psychiatry, Psychology and Neuroscience, King's College London, London, UK

¹⁶Stress, Psychiatry and Immunology Lab & Perinatal Psychiatry, Maurice Wohl Clinical Neuroscience Institute, King's College London, London, UK

¹⁷Institute of Clinical Sciences, University of Birmingham, Birmingham, UK

^{*}Corresponding author bill.deakin@manchester.ac.uk †In memoriam

Declared competing interests of authors: Bill Deakin reports grants from P1vital Ltd (Wallingford, UK) and grants and personal fees from Autifony Therapeutics Ltd (Stevenage, UK) outside the submitted work; he contributed financial support to the study from his National Institute of Health Research (NIHR) Senior Investigator award, and he became a member of the NIHR Efficacy and Mechanism Evaluation Panel after completion of the study. John Suckling reports non-financial support from Cambridge NIHR Biomedical Research Centre; non-financial support from Behavioural and Clinical Neuroscience Institute, University of Cambridge; and grants from Cambridgeshire and Peterborough NHS Trust Research Strategy Committee during the conduct of the study. Stephen M Lawrie reports that he has received personal fees from Otsuka Pharmaceuticals (UK) Ltd (Wexham, UK) and Sunovion Pharmaceuticals Inc. (Marlborough, MA, USA), and personal fees and research support from Janssen-Cilag Ltd (High Wycombe, UK). Rachel Upthegrove reports personal fees from Sunovion Pharmaceuticals, Inc., outside the submitted work. Nusrat Husain is chairperson of the board of trustees of Manchester Global Foundation, a not-for-profit organisation to address inequalities and promote health and well-being within the UK and globally. Peter Jones reports personal fees from Member Scientific Advisory Board – Janssen-Cilag Ltd and personal fees from Member Scientific Advisory Board – Ricordati Pharmaceuticals Ltd (Reading, UK) outside the submitted work. Danuta Lisiecka-Ford reports grants from NIHR during the conduct of the study. Shôn Lewis reports personal fees from Affigo.io (Manchester, UK) and Xenzone (Manchester, UK) outside the submitted work. Thomas Barnes reports personal fees from Sunovion Pharmaceuticals, Inc., Lundbeck Ltd (St Albans, UK), Newron Pharmaceuticals SpA (Bresso, Italy) and Janssen-Cilag Ltd outside the submitted work. Steven CR Williams reports research funding from Bionomics Ltd (Thebarton, SA, Australia), Eli Lilly and Company (Indianapolis, IN, USA), the Engineering and Physical Sciences Research Council, GlaxoSmithKline plc (Middlesex, UK), Johnson & Johnson (New Brunswick, NJ, USA), Lundbeck Ltd, NIHR, Pfizer Inc. (New York, NY, USA), Takeda Pharmaceutical Company Ltd (Tokyo, Japan) and the Wellcome Trust. Carmine M Pariante reports that, in the last 5 years, he has received research funding from Johnson & Johnson, a pharmaceutical company interested in the development of anti-inflammatory medications for use in psychiatry, and from Medical Research Council- and Wellcome Trust-funded research consortia that also include GlaxoSmithKline plc, Johnson & Johnson and Lundbeck Ltd; however, the work in this publication is completely independent from this funding. Richard Drake reports honoraria for presentations at meetings and advisory board membership sponsored by Lundbeck Ltd, Janssen-Cilag Ltd (High Wycombe, UK), Sunovion Pharmaceuticals, Inc., and Otsuka Pharmaceuticals (UK) Ltd that have been paid to the University of Manchester, and the subjects of these meetings do not represent a conflict of interest.

Published August 2019 DOI: 10.3310/eme06070

Plain English summary

The BeneMin RCT

Efficacy and Mechanism Evaluation 2019; Vol. 6: No. 7

DOI: 10.3310/eme06070

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

We have completed a clinical trial in 207 people who had received a diagnosis of schizophrenia in the previous 5 years. These participants agreed to add a well-known antibiotic called minocycline to their usual treatment for 1 year. The aim was to check previous evidence that minocycline may improve symptoms of schizophrenia such as self-neglect, social withdrawal, apathy and unemployment (so-called negative symptoms) that can develop and persist after treatment with current antipsychotic medication. We also tested the idea that minocycline might work by supporting the functioning of brain cells, perhaps through its anti-inflammatory actions.

Interested patients in early intervention and community services in 11 UK NHS trusts were taken through the study by local research teams. Neither the participants nor the researchers knew who was taking real minocycline and who was taking matching dummy (placebo) capsules. Clinical interviews and questionnaires were used to follow any changes in psychotic and negative symptoms as well as general health and social functioning over the 12 months of treatment. Brain functioning was assessed using magnetic resonance imaging scans and performance tests. Molecules associated with inflammation called cytokines were measured in blood samples.

When we decoded the results after the last participant had completed the study, we found that there were no differences in any of the clinical measures between those who had taken minocycline and those who had taken the placebo capsules. Further analyses are under way, but we found that there was no general tendency to worsening of symptoms, brain function or inflammation over the 12 months of standard care and that taking minocycline provided no additional benefit. The results make it unlikely that there is an inflammatory process that gives rise to negative symptoms in recent schizophrenia that is modifiable by minocycline. We cannot exclude the possibility that minocycline would benefit a more severely ill group who have not responded to standard treatments.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme was set up in 2008 as part of the National Institute for Health Research (NIHR) and the Medical Research Council (MRC) coordinated strategy for clinical trials. The EME programme is broadly aimed at supporting 'science driven' studies with an expectation of substantial health gain and aims to support excellent clinical science with an ultimate view to improving health or patient care.

Its remit includes evaluations of new treatments, including therapeutics (small molecule and biologic), psychological interventions, public health, diagnostics and medical devices. Treatments or interventions intended to prevent disease are also included.

The EME programme supports laboratory based or similar studies that are embedded within the main study if relevant to the remit of the EME programme. Studies that use validated surrogate markers as indicators of health outcome are also considered.

For more information about the EME programme please visit the website: http://www.nets.nihr.ac.uk/programmes/eme

This report

The research reported in this issue of the journal was funded by the EME programme as project number 09/100/23. The contractual start date was in July 2011. The final report began editorial review in July 2017 and was accepted for publication in March 2019. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, NETSCC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the EME programme or the Department of Health and Social Care.

© Queen's Printer and Controller of HMSO 2019. This work was produced by Deakin et al. under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This issue may be freely reproduced for the purposes of private research and study and extracts (or indeed, the full report) may be included in professional journals provided that suitable acknowledgement is made and the reproduction is not associated with any form of advertising. Applications for commercial reproduction should be addressed to: NIHR Journals Library, National Institute for Health Research, Evaluation, Trials and Studies Coordinating Centre, Alpha House, University of Southampton Science Park, Southampton SO16 7NS, UK.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Honorary Professor, University of Manchester, and Senior Clinical Researcher and Associate Professor, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HS&DR, PGfAR, PHR journals) and Editor-in-Chief of HS&DR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Senior Scientific Advisor, Wessex Institute, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Director, NIHR Dissemination Centre, UK

Dr Catriona McDaid Senior Research Fellow, York Trials Unit, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Professor of Wellbeing Research, University of Winchester, UK

Professor John Norrie Chair in Medical Statistics, University of Edinburgh, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, UCL Great Ormond Street Institute of Child Health, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Professor Martin Underwood Warwick Clinical Trials Unit, Warwick Medical School, University of Warwick, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk